Author’s response to reviews

Title: Challenges in Defining the Rates of ADHD Diagnosis and Treatment: Trends over the Last Decade

Authors:

Michael Davidovitch (davidom@netvision.net.il)
Gideon Koren (koren_gid@mac.org.il)
Naama Fund (fund_n@mac.org.il)
Maayan Shrem (Maayan.shrem@mail.huji.ac.il)
Avi Porath (porath.avi@mac.org.il)

Version: 3 Date: 20 Oct 2017

Author’s response to reviews:

12/10/2017

Paramala Santosh,

MBBS, Dip NB (Psych), MD, PhD, FRCPsych

BMC Pediatrics

Dear Prof. Santos

Thank you for considering this article for publication.

We have addressed all of the editor comments below.
1. Please include a list of abbreviations used in the manuscript before the Declarations section – Done.

2. Please include reference number for ethics approval – Done.

3. Please ensure author contributions listed in the authors contribution section are reflective of ICMJE guidelines – Done

4. In the Author contributions section, the sentence "All authors have read and approved the final version of this manuscript" should also be included – Done

5. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References – Done.

6. Please upload a clean version of the manuscript without tracked changes – Done.

We hope these changes are satisfactory and look forward to your response.

Sincerely,

Michael Davidovitch MD
Child Neurology and Development
Head of Child Development and ADHD Services
Maccabi Healthcare Services
Chairman of the Israeli Society of Developmental Pediatrics (Israel Medical Association)